Author's response to reviews

Title: Management of major bile duct injury after laparoscopic cholecystectomy: A case report and review of the literature

Authors:

Andreas Manouras (amanouras@hippocratio.gr)
Nikolaos Pararas (npararas@hotmail.com)
Pantelis Antonakis (antonakispantelis@yahoo.gr)
Emanuel E. Lagoudianakis (redemlag@yahoo.gr)
George Papageorgiou (awace@yahoo.gr)
Ioannis G. Dalianoudis (johndalas@gmail.com)
Manousos M. Konstadoulakis (lbsures@hippocratio.gr)
Vaggelogiannis Katergianakis (vkater@hotmail.com)

Version: 5 Date: 11 September 2008

Author's response to reviews:

According to the reviewer's we have made changes to the article in order to conform with the suggestions that they have made. We have add a flow chart showing the diagnostic procedure that a patient, with suspected bile duct injury, undergo. We are discussing what other surgical options we have for the treatment of the patient and finally the postoperative management of the anastomosis and how long should be the biliary tubes be in place.